Author's response to reviews

Title: Recruitment of Young Adults into a Randomized Controlled Trial of Weight Gain Prevention: Message Development, Methods, and Cost

Authors:

Deborah F Tate (dtate@unc.edu)
LaRose G Jessica (jlarose@vcu.edu)
Leah P Griffin (lgriffin@wakehealth.edu)
Karen E Erickson (keericks@email.unc.edu)
Erica Ferguson (eferguson@lifespan.org)
Letitia Perdue (lperdue@wakehealth.edu)
Mark Espeland (mespelan@wakehealth.edu)
Rena R Wing (rwing@lifespan.org)

Version: 4 Date: 30 July 2014

Author's response to reviews:

An Acknowledgements section has been added to attest that the authors do not require permission to re-publish images of the recruitment materials.